Contents lists available at ScienceDirect

Energy Policy

journal homepage: www.elsevier.com/locate/enpol

How long does innovation and commercialisation in the energy sectors take? Historical case studies of the timescale from invention to widespread commercialisation in energy supply and end use technology



ENERGY POLICY

Robert Gross^a, Richard Hanna^{a,*}, Ajay Gambhir^b, Philip Heptonstall^a, Jamie Speirs^c

^a Imperial College Centre for Energy Policy and Technology, Imperial College London, 16-18 Prince's Gardens, South Kensington, London SW7 1NE, UK

^b Grantham Institute, Imperial College London, Exhibition Road, South Kensington, London SW7 2AZ, UK

^c Sustainable Gas Institute, Imperial College London, 10 Prince's Gardens, South Kensington, London SW7 1NA, UK

ARTICLE INFO

Keywords: Technology commercialisation Innovation policy Low carbon innovation Energy technology diffusion Mission Innovation Innovation timescales

ABSTRACT

Recent climate change initiatives, such as 'Mission Innovation' launched alongside the Paris Agreement in 2015, urge redoubled research into innovative low carbon technologies. However, climate change is an urgent problem – emissions reductions must take place rapidly throughout the coming decades. This raises an important question: how long might it take for individual technologies to emerge from research, find market opportunities and make a tangible impact on emissions reductions? Here, we consider historical evidence for the time a range of energy supply and energy end-use technologies have taken to emerge from invention, diffuse into the market and reach widespread deployment. We find considerable variation, from 20 to almost 70 years. Our findings suggest that the time needed for new technologies to achieve widespread deployment should not be overlooked, and that innovation policy should focus on accelerating the deployment of existing technologies as well as research into new ones.

1. Introduction

The role and importance of technological innovation in reducing greenhouse gas emissions is well established in national and international policies (DECC, 2012; CCC, 2013, IPCC, 2015, IEA, 2015). Recent initiatives aimed at accelerating innovation in low carbon energy technologies focus in particular on enhancing government funding for research, development and demonstration (RD&D) (Mission Innovation, 2016; Breakthrough Energy Coalition, 2016; King et al., 2015; Dechezleprêtre et al., 2016). Yet if low carbon technologies are to play a substantial role in reducing carbon emissions in the coming decades, then it will be necessary to not just research, develop and demonstrate them, but to also make them commercially available and deploy them at scale, since emissions must fall rapidly during the period to 2050 to meet internationally agreed climate targets (IPCC, 2015, UNFCCC, 2015).

Much of the substantial literature on 'innovation systems' recognises that innovation policy needs to include *both* increased funding for RD& D *and* targeted measures to create market opportunities for low carbon technologies (IEA, 2000; Anderson et al., 2001; Foxon et al., 2005; Gross et al., 2012; Winskel et al., 2011), with ongoing debate on the optimal mix for specific technologies (Helm, 2010, 2017; Nemet and Baker, 2009; King et al., 2015; Policy Exchange, 2011).

However, the amount of *time* required for new technologies to emerge from fundamental research, go through demonstration and early stage deployment and diffuse into the market place also matters greatly, for the obvious reason that policy makers and innovators need a sense of how rapidly such technologies can make a material impact on reducing emissions. For this reason mitigation scenarios produced by integrated assessment models, which are central to low-carbon pathways analysis, are increasingly scrutinised with regard to their realworld feasibility, often by comparing their rates of low carbon energy technology deployment to historical rates of deployment of existing energy technologies (Wilson et al., 2013; Iyer et al., 2015; Napp et al., 2017; Van Der Zwaan et al., 2013, van Sluisveld et al., 2015), so as to determine whether or not they are simply "computerised fairy tales" (Smil, 2010b).

The issue of the time taken for technologies to commercialise has received relatively little attention in the innovation literature, in spite of its criticality to understanding the feasibility of future mitigation pathways and directing technology innovation and deployment policy. As discussed in Section 2.2, there have been a number of analyses on

* Corresponding author.

E-mail address: r.hanna@imperial.ac.uk (R. Hanna).

https://doi.org/10.1016/j.enpol.2018.08.061

Received 8 March 2018; Received in revised form 10 July 2018; Accepted 27 August 2018 Available online 10 October 2018

0301-4215/ © 2018 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/BY/4.0/).

the timescale for the growth of different energy sources and energy technologies from initial prototype to various stages of development and maturity. But there has not yet been a detailed analysis of the timescale from invention to an agreed definition of widespread commercialisation of energy technologies. A key contribution of this paper is to provide new empirical evidence and insights on the topic of commercialisation timescales.

As we explain further in Section 2, there is a multitude of definitions and conceptualisations of various innovation stages. The paper therefore proposes new definitions of different stages in technology development and deployment that are designed to be simple and readily intelligible to non-specialists. In particular, the paper develops a new definition of 'widespread commercialisation' that represents a level of deployment of a technology that can be considered fully commercialised, but with potential to continue to increase market share. This allows innovation timescales to be presented in an accessible form that permits comparison between technologies and can be readily used by policy makers who need to understand how long it could take for new low-carbon technologies to become widely commercialised and able to make a material contribution to emission abatement.

To summarise, the paper's contribution is to enhance knowledge on innovation timescales, by providing empirical evidence and commentary on how long it has taken selected case study technologies to emerge from RD&D and achieve a readily understandable level of widespread commercialisation.

The rest of this paper is set out as follows: Section 2 presents a brief background on innovation processes and frameworks, before discussing the recent literature that specifically examines the timescales and rates of energy technology penetration; Section 3 describes the methodology used to calculate the innovation timescales for a range of energy supply and end use technologies, as well as justifying the selection of these technology case studies; Section 4 presents and discusses the results; Section 5 discusses the findings and limitations of the study. Section 6 concludes and discusses policy implications.

2. Background: the literature on innovation systems, stages and timelines

This section firstly provides a brief overview of the frameworks used to describe the stages and processes involved in technological innovation and deployment. Empirical analyses of energy transitions that have examined the timescales over which these stages and processes have occurred are then summarised. Finally, the section identifies gaps in the existing literature with regard to innovation and widespread commercialisation timescales, and discusses the contribution that this study makes.

2.1. Frameworks and models of energy technology innovation and deployment

There is a large literature on technology innovation in energy and other sectors. Early perspectives focused on a relatively simple, onedirectional journey from basic research to applied research to technology development and diffusion, suggesting that the optimal way to increase the output of new technologies was to put more resources into R&D, a process called technology or supply-push (Schumpeter, 1934). An alternative perspective, demand-pull, gained traction in the 1950s (Carter and Williams, 1957), arguing that demand for products and services was more important in stimulating inventive activity than advances in the state of knowledge (Allen, 1967). Fri (2003) contends that the model of innovation called research, development, demonstration and deployment (RDD&D), which combines supply-push and demand-pull activities has set the form for virtually all discussions on energy innovation.



Fig. 1. Typical technology S-curve. Source: Taylor and Taylor (2012)

There are a variety of models of technology diffusion in the literature which attempt to explain factors governing the speed of adoption of new technologies and giving rise to the typical shape of the technology S-curve (Geroski, 2000), illustrated in Fig. 1. Diffusion of innovations theory (Rogers, 1962) sets out the conditions under which innovative products may become accepted by consumers over time, so as to lead to their widespread acceptance and purchase. Whether consumers purchase an innovative product depends on how much they are aware of any relative advantage over alternative products, and whether they are motivated to find out more about the innovation (Faiers and Neame, 2006).

Five groups of adopters are identified by Rogers (1962): innovators, early adopters, the early majority, the late majority, and laggards. The Bass model of product growth (Bass, 1969) builds on Roger's adoption groups, so that early adopters through to laggards are considered to be 'imitators' of initial innovators. As more consumers adopt a product, imitators are influenced in the timing of their adoption by increasing social pressure to take up a product. According to the Bass model, the probability that the initial purchase of a new product will be made at a given point in time is 'a linear function of the number of previous buyers' (Bass, 1969).

The 'Epidemic' model (Bartholomew, 1973) is commonly used to account for the S-curve, and is based on the assumption that a lack of available information about a technology constrains its rate of uptake. Alternatively, the 'Probit' model (Davies, 1979), assumes that different firms have different objectives and skills, and therefore do not all adopt a technological innovation at the same time. In this model, diffusion takes place as different types of firms choose to adopt a new technology (Geroski, 2000).

The RDD&D, technology lifecycle and technology diffusion models present a somewhat linear, successive picture of technology development, which has been challenged by recent approaches which have noted the importance of more complex, systemic feedbacks between the supply and demand sides (Foxon, 2003), as well as the role of agents and actors in developing and deploying technologies within a broader socio-technical landscape. Examples of specific approaches include 'technological innovation systems', 'technological transitions', and the 'multi-level perspective' (Foxon, 2003). Technological Innovation Systems (TIS) theory aims to understand how new technologies can evolve through interactions between actors, networks and institutions (Bergek et al., 2008; Bento and Fontes, 2015). Transitions theory emphasises the importance of technological and market niches by which an innovation can be protected from normal market conditions and nurtured for a period of time. A key element of transitions theory is the 'multi-level perspective' which stresses that transitions do not only involve changes in technologies, but also changes in user practices, regulation, industrial networks, infrastructure and symbolic meaning or culture (Geels, 2002).

The innovation frameworks and models of technology diffusion so far discussed highlight the importance of different activities to develop and deploy technologies, the institutions and actors involved in their penetration into the energy system, the role of consumer behaviour and information diffusion, as well as the different stages of technology maturity. However, the above concepts do not explicitly allow an *ex ante* quantification of the time taken for these processes to occur. To address this question, a more empirical literature has emerged, as discussed in the following sub-section.

2.2. Innovation and deployment in the energy sector

The foci of existing empirical analyses on innovation timescales are diverse, ranging from analysis of discrete technologies moving through a particular innovation stage (Bento and Wilson, 2016) to analysis of long term historical transitions of fuel sources, energy services and industries (Fouquet, 2010; Perez, 2002; Smil, 2010b; Pearson and Arapostathis, 2017). In addition, a very recent literature seeks to understand the feasibility of future low carbon scenarios by comparing low carbon energy technology deployment rates with historical deployment rates for existing energy technologies (Wilson et al., 2013; Iyer et al., 2015; Napp et al., 2017; Van Der Zwaan et al., 2013, van Sluisveld et al., 2015).

It has been suggested that technological innovations typically take between five and seven decades to travel from invention to significant market shares (O'Neill et al., 2003, Grubler, 1998). Perez (2002) considers 'radical' innovations from the 1770s to 2000s, and observes that successive 'technological revolutions' took between 43 and 66 years to reach maturity. These include innovations associated with the industrial revolution, steam railways, steel and electricity and automobiles. Wilson and Grubler (2010) cite a variety of examples and conclude that on a global scale it has taken 80–130 years for new energy technology clusters to achieve market dominance, and about twice as long when considering the entire technology life cycle from first introduction to market maturity.

However, the timescales needed for new technologies to diffuse widely are contested. For example, Sovacool (2016) argues against the widespread view that energy transitions are inevitably slow, noting that it runs contrary to a number of important 'quick' empirical examples, from nuclear power in France to improved cooking stoves in rural China (Sovacool, 2016). Further rapid transitions include from petrol to biofuels in cars in Brazil (Grad, 2006), the switch from coal-based town gas to natural gas in the UK (Arapostathis et al., 2013) and the rapid expansion of wind energy in Denmark (Napp et al., 2017).

However there is no consensus definition of technological maturity in the sense of when or at what market share a particular technology can be considered as firmly established, widely available and commercially viable. Kramer and Haigh (2009) identify a distinct stage at which energy sources reach what they term "materiality", whereby they constitute about 1% of global primary energy supply, at which point their annual growth transitions from exponential to linear. Wilson et al. (2013) focus on the time taken for a range of energy technologies to grow from 10% to 90% of their long-term saturation level, termed the "duration" level by the authors, and in most cases taking several decades. One reason for the apparent disagreements over timescales would appear to be a lack of agreement over what constitutes widespread deployment.

Grubler et al. (2016) elaborate on differences between slow and rapid energy transitions, arguing that the examples of quick transitions provided by Sovacool (2016) are generally either new technologies which substitute for older ones or have been previously used in other markets, or which offer a high degree of tangible benefits for adopters. Conversely, slower energy transitions may require changes to take place in multiple technologies, infrastructures and institutions, extensive time to develop and test new concepts, and may entail investments in largescale and costly technologies and infrastructures whose benefits may be longer term, social or environmental (i.e. non-market), and therefore less immediately tangible to consumers (Grubler et al., 2016).

Some of the most rapid energy technology growth rates have been used as benchmarks against which to test the feasibility of low carbon transitions. For example, Iver et al. (2015) survey a range of national and international historically achieved energy technology deployment rates to inform a maximum 15% annual growth rate constraint on individual energy technologies in their low carbon pathway scenarios. This compares to Kramer and Haigh (2009) who observe an approximate 25% annual growth rate in primary energy sources achieved over the early decades of their deployment. Wilson et al. (2013) use a different combination of metrics to compare future low carbon technology deployment rates to historical rates, based on the time taken to grow from 10% to 90% of their eventual market saturation level. van Sluisveld et al. (2015) further develop this analysis to account for the growing size of the economy and energy system over time. Napp et al. (2017) combine a number of historical technology deployment metrics to impose constraints on modelled low-carbon pathways, concluding that if all available historical energy system transition metrics are taken into account, then the ability of their energy system model to meet a 2 °C climate target is severely compromised.

Overall it appears that there is conflicting evidence on the speed of energy technology penetration into energy systems, and considerable variability in the metrics used to measure the timescale to widespread deployment.

2.3. Gaps in the literature and contribution of this study

This paper sets out to improve understanding and clarity for policy makers on how long it can take for new technologies to significantly penetrate into the energy system from their point of invention. The paper addresses a number of gaps in the existing literature. In particular there are a wide range of metrics used to assess timescales in the existing innovation literature, and a lack of formal definitions of stages of technology development, particularly with respect to when a technology can be considered to be widely commercialised (Bento and Wilson, 2016; Grübler et al., 1999; Lund, 2006; Rogers, 1962; Smil, 2010a; Wonglimpiyarat, 2005). There are also a lack of analyses charting the whole innovation journey from invention to maturity. Bento and Wilson (2016) use a variety of metrics to measure the length of the 'formative phase', which represents the point at which a technology transitions from an emergent market into a more established market. Our paper goes beyond this analysis in applying a clear and comparable definition of widespread use that charts a more extended journey towards wider commercialisation.

We provide further weight to empirical evidence on historical deployment rates of energy supply and end use technologies which can be compared with deployment rates assumed in integrated assessment models used to construct low carbon scenarios. In general our findings contribute to existing empirical studies which contest whether diffusion timescales for energy technologies are longer and may take place over several decades (Wilson et al., 2013; Grubler et al., 2016) or more rapidly for particular technology substitutes (Sovacool, 2016). One surprising result from our analysis is the extended time taken for the car to reach widespread commercialisation, which is related to a number of contextual, income and economic growth-related factors (see Section 5.1).



Fig. 2. Schematic illustrating approach to selecting case study technologies through the evidence review.

3. Methodology

This paper has two broad areas of investigation: an empirical review of the time taken from invention to widespread commercial use for a range of technologies and; a conceptual and definitional discussion of innovation stages, rooted in the literature on innovation. The empirical dimension of the paper, as discussed in Section 3.1, is based upon an evidence review undertaken using an approach developed by the authors for the UK Energy Research Centre (Speirs et al., 2015). The conceptual discussion of innovation stages, as presented in Section 3.2, is undertaken in order to develop our own definitions of innovation start and end points, which we apply to the empirical case studies.

3.1. Evidence review on innovation

The evidence review sought out a range of technologies and used a variety of innovation related search words. Relevant literature was identified through keyword searches for innovation concepts, descriptors of time and technologies or products (Appendix A) within two databases, Elsevier Science Direct (Elsevier, 2015) and World Cat (OCLC, 2015). Initially, innovation and temporal keywords were used as the search terms, and subsequently combined with product or technology keywords based on the initial searches (Fig. 2). A snowballing technique (Greenhalgh and Peacock, 2005) was also employed which involved following citation trails of documents identified through the keyword searches and including relevant cited documents in the evidence database, in order to widen the number of potential technology or product case studies. Ultimately, 13 successfully commercialised innovations were identified as being instructive to the research aims and for which sufficient information was available to quantify timescales from invention to widespread commercialisation (Fig. 2).

Following the filtering of retained search results key descriptive information on each of the results were captured: (i) the innovative product or technology considered; (ii) the timescales for specific innovation stages presented and; (iii) the geographic region (if not global). It should be noted that there are several other technologies for which innovation journeys have been examined and discussed in the literature. These include: various innovations in the US steel industry (Gold et al., 1984); the dieselisation of railways (Mansfield, 1965); the diffusion of the basic oxygen furnace (Oster, 1982); and alternative fuelled vehicles (Yeh, 2007). However, as with Lund (2006), which compares market penetration rates for 11 energy technologies, we limit ourselves to those case studies for which data availability permitted a like-for-like comparison of innovation timescales.

We compare a wide diversity of technologies – energy supply and end use technologies, component parts (batteries), and various 'assembled products' (Utterback, 1994) including consumer electronics and the car. This technology diversity allows us to compare the time taken to widespread commercialisation for both technological innovations which can help to achieve energy systems decarbonisation, and wider consumer products or technologies such as the car, mobile phone, VCR and ATMs/cash cards.

Through this comparison we can observe differences in innovation timescales between technologies aimed at climate change mitigation and those which have experienced strong market demand from consumers or provide distinct novel or replacement services for consumers. The 13 innovations selected for the construction of innovation timelines were organised into three groups: novel energy end use and consumer products (new products with entirely new markets); replacement energy end use and consumer products (new technologies which replace but perform a similar function to an existing product with an incumbent market); and electricity generation technologies.

We deliberately chose technologies which have entered widespread use and have been commercially successful and widely adopted in at least some markets, and do not consider 'failed' technologies. This aspect of technology choice was largely pragmatic, reflecting a choice of technologies for which good data were available and which would be widely recognised by policymakers and a wider audience as being in widespread use, though not necessarily ubiquitous. We included electricity generation technologies that might be viewed as less than fully commercialised (for example wind power or solar PV) or for which market size might have been larger (nuclear power). The judgements

Table 1

Innovations and geographies.

Innovation category	Product or technology	Year of invention	Geography/ geographies applying to market introduction and widespread commercialisation
Novel products for new markets – energy end	Cars	1885	US
use and consumer products	Cathode Ray Tube (CRT) TV	1912	US
	ATM/Cash cards	1964	UK
	Video cassette recorder (VCR)	1956	UK
Replacement products – energy end use and	Mobile phone	1970	US
consumer products	Thin Film Transistor Liquid Crystal	1988	Global
	Display (TFT-LCD) TV		
	Lithium ion rechargeable batteries for consumer products	1979	Global
	Compact fluorescent light bulbs	1976	UK
	(CFLs)		
	LED lights	1992	UK
Electricity generation technologies	Combined cycle gas turbines (CCGT)	1949	UK
	Wind electricity	1957	Denmark
	Nuclear power	1941	France
	Solar photovoltaics (solar PV) – grid- connected	1954	Germany

we made around definitions of widespread commercialisation and overall market share are discussed in Section 3.2.

The categories are chosen to help provide insight into some of the key factors that affect how rapidly new products reach commercial maturity. As we discuss below, the innovation literature suggests that replacement products may reach widespread commercialisation more quickly than those which require the creation of new markets, infrastructures, regulatory environment and consumption patterns (Grübler et al., 1999). The distinctions are approximate in some instances and there may be debate at the margin, for example whether cars ('horseless carriages') are replacement horses, or whether VCRs could be considered a replacement for reel to reel cinematography. We argue that the capabilities of cars and VCRs represent such fundamental departures from any predecessors that it is better to consider them as entirely new products.

We note that Saviotti and Metcalfe (1984) discuss a categorisation of products in terms of whether they offer new technical characteristics and/or new services. In this regard, we consider that new forms of electricity generation are also replacement technologies – since they provide the same service or output as the power generation technologies which they replace. We give separate attention to electricity generation technologies because many analyses suggest that electricity sector decarbonisation is of particular importance in reducing overall emissions (IEA, 2015, IPCC, 2015, CCC, 2015, Bassi et al., 2013) and in addition the market for these technologies is not end users (as in the case of the other technologies), but rather utilities providing electricity supply.

The geography chosen to assess the time taken for innovation is also important, since in many instances particular countries or regions provided important early markets¹ for new technologies. In a manner similar to that used by Bento and Wilson (2014), we assess the time taken to reach widespread commercialisation in early or lead markets at a national scale. In two cases market diffusion occurred rapidly across multiple countries over a short period of time (LCD-TVs and lithium ion rechargeable batteries) and we therefore present data for global market growth for these technologies. Since the point of invention is not geographically restricted and may be distinct from the country that went on to provide the first market for an emerging technology we take the year of invention as the time it occurred in any location in the world.

Table 1 provides an overview of the main technologies and their year of invention, technology categories and geographies included in the analysis.

3.2. Defining the different stages of the innovation journey

As shown in Table 2, the innovation literature uses a wide range of terminology to describe progressive stages of technological innovation (Dismukes et al., 2009; Wonglimpiyarat, 2005; Perez, 2002; Taylor and Taylor, 2012; Gao et al., 2013; Yeo et al., 2015). However, it is possible to identify phases for which multiple terms are used but which appear to be, in broad terms, consistent. For the purposes of this study, the timescales for each innovation were initially grouped according to three progressive phases in the innovation life cycle, identified for every product in the evidence assessment: 'development', 'market formation', and 'growth and diffusion'. These categories were further rationalised into two composite phases: 'invention, development and demonstration' and 'market deployment and commercialisation' (Fig. 3). The first of these begins with invention and encompasses research and development, and ends at the point of market introduction of an innovation. The second phase starts with market introduction and ends when a technology or product reaches a point which we have defined as 'widespread commercialisation'.

Fig. 3 provides a schematic of the phases defined for this study, and the wider terminology used to define different phases in the literature. Defining the start and end points of these innovation stages is not uncomplicated and providing appropriate definitions and distinctions therefore comprised an important component of the research.

We have generally taken the point of invention for each innovation as the year in which a product or a technological application was conceived and tested at laboratory scale – the first laboratory test of a prototype such as a solar cell or cathode ray tube. This is similar to the narrow definition of innovation in Grübler et al. (1999) as the first 'practical application of an invention'. However for many of the innovations studied, the basic scientific or engineering principles underpinning an innovation predate even the laboratory stage by several decades. For example, in 1954 Bell Laboratories developed a siliconbased, laboratory-scale solar PV cell which could be translated rapidly into practical application. In the following year, Bell Laboratories used this breakthrough cell design to construct a silicon solar PV module for outdoor use to power telephone lines in Georgia. Yet, the photovoltaic effect itself was established using selenium in the late 1870s (Perlin, 1999).

¹ These are not necessarily the definitive lead markets for each technology included in this study, and they were also selected based on available data permitting timescales to be quantified according to the definitions for innovation stages applied in the analysis.

Key term	Definition
Invention	Origination of an idea as a technological solution to a perceived problem or need
Innovation	Putting ideas into practice through an iterative process of design, testing, application and improvement
Research and development (R&D)	Knowledge generation by directed activities (e.g. evaluation, screening, research) aimed at developing new or improving on existing technological knowledge
Demonstration	Construction of prototypes or pilots for testing and demonstrating technological feasibility and/or commercial viability
Research, development and demonstration (RD&D)	A commonly used grouping of the main pre-commercial stages of the innovation cycle
Niche markets	Application of a technology in a limited market setting (or niche) based on a specific relative performance advantage (or on public policy incentives) and typically protected in some way from full market competition
Market formation	Activities designed to create, enhance, or exploit niche markets and the early commercialisation of technologies in wider markets
Diffusion/deployment	Widespread uptake of an energy technology throughout the market of potential adopters
Innovation, technology or product (development) life cycle	The sequence of processes and stages of an innovation's journey from invention right through to senescence or obsolescence

ay technology innegation (adapted from Wilson and Crubler 2014)



Fig. 3. Phases of the innovation timeline (authors' own).

For all end use and consumer products, we define market introduction as the point in time when products first became commercially available. For electricity generation technologies, market introduction refers to the commissioning of the first commercial units which generated electricity for supply to the grid, discounting early demonstration installations or grid-connected prototypes – similar to the approach taken by McDowall (2015) to pinpoint market entry for hydrogen fuel cell vehicles. Details of the judgements made for each technology are provided in Appendix C.

Defining 'widespread commercialisation' is more challenging. This is because a judgement is needed about when a technology is 'widespread' or 'mature' or 'established'. The term we use throughout this paper is 'widespread commercialisation'. This is not as hard and fast a concept as invention point or the entry of the first products into a market. We therefore utilised and extended definitions from the literature, which are summarised in Table 3. For example, Grübler et al. (1999) refer to the notion of the 'pervasive diffusion' of technologies in which a rapidly increasing commercial market share of 5–50% is achieved. Based on Rogers (1962), wider commercialisation might be represented by between 15% and 50% of potential adopters taking up a new product: if the early majority adopts an innovation, this would cover half the eventual number of adopters, as opposed to the first 15% being accounted for by initial innovators (2.5%) and early adopters (12.5%). Lund's comparison of market penetration rates for energy technologies uses a definition of 'market potential' taken at different market shares (1-50%) for different technologies, varying both by constraints to reaching higher market shares (e.g. renewable energy intermittency) and the geographical size of the market (Lund, 2006). Vaclav Smil provides a more consistent definition - referring to the time taken for the rise of a new "fuel or prime mover" from a 5% to a 25% share of global energy supply (Smil, 2010a). Wonglimpiyarat (2005) refers to a 'means to market' phase which begins after: '...the distribution capabilities (distribution channels) are sufficient to access 20% of the target population of users for the innovation'. Bento and Wilson propose that a 10% market share represents one indicator of the end of the 'formative phase' (Bento and Wilson, 2016). The precise meaning of the end of the formative phase is linked to innovation systems theory, representing a transition point between a technology that is in a nascent market and one that is in a more established market. In terms of the journey towards wider commercialisation, this falls some way short of the definition of widespread use sought for this paper.

There are thus no agreed quantitative definitions of invention, market introduction or widespread commercialisation in the literature. The size and specification of market share that is chosen to represent widespread commercialisation differs considerably within the literature and there is no more consistency (and indeed some ambiguity) over what reference point – ultimate market size, fully adopted, saturated, etc. – is used (Table 3). The definitions of start points and – even more challenging – the end point used herein were therefore chosen to permit technologies to be compared on a 'like for like' basis within each of the technology categories described in the main text. They were created to

Table 3

Definitions of widespread market diffusion of innovations: examples from the literatur
--

Author(s) (year)	Name / description of metric	Definition of widespread market diffusion
Bento and Wilson (2016)	Formative phase	10% of eventual market saturation (units produced and capacity installed); or 10% of maximum unit capacity
Grübler et al. (1999)	Pervasive diffusion	5-50% commercial market share
Lund (2006)	Market potential	1-50% market potential or adoption ceiling
Rogers (1962)	Early majority	15-50% of potential adopters taking up an innovation
Smil (2010a)	Time for a new 'fuel or prime mover' to achieve given shares of total global energy supply	25% share of global energy supply
Wonglimpiyarat (2005)	Means to market	Distribution channels can access 20% of target population of users for the innovation

Table 4

Defining the point of widespread commercialisation.

0 1 1		
Innovation category	How widespread commercialisation is defined in this study	Geographies applicable
Novel products for new markets	When the yearly number of products in ownership, use or installed in the marketplace first reaches 20% of peak products in use (i.e. the maximum number of products in ownership, use or installed in the marketplace, based on annual time series data).	UK, US
Replacement products or technologies	As above	UK, US, Global
Electricity generation technologies	When installed capacity first reaches 20% of peak installed capacity (i.e. the maximum installed capacity of that technology based on annual time series data).	UK, Denmark, France, Germany

help inform policy by providing a reasonable approximation of the time taken for a successfully commercialising product to emerge from the research stage and commercialise – in particular, to represent when use of new products became widespread, though not necessarily ubiquitous or market saturated. This is important because a principal goal of the research is to assess the time needed to achieve usage on a scale where it might be material to carbon abatement goals. Whilst acknowledging that there is no 'right answer' (it could be 15% or even 50% of ultimate market) we propose 20% of ultimate market size as an approximate measure of widespread commercialisation, for all categories of technology. More precisely, our definitions are as follows (see also Table 4):

We consider both *novel products for new markets* and *replacement products* to have reached widespread commercialisation when the number of products in use (i.e. the yearly number of products in ownership, use or installed in the marketplace) first reaches 20% of peak products in use (i.e. the maximum number of products in ownership, use or installed in the marketplace, based on annual time series data). *Electricity generation technologies* are considered to be in widespread commercialisation when their installed capacity first reaches 20% of peak installed capacity of the technology in question (the maximum installed capacity of that technology – not all electricity capacity – based on annual time series data).

Only a few of the technologies included in this analysis are obsolete (e.g. CRT TVs have been replaced by LCD TVs; VCRs were replaced by DVD players), while the installed capacity of nuclear power in France peaked in 1999. Most of the innovations included in the study are contemporary, commercially established technologies, with sales that continue to add to the cumulative stock of products, and many have yet to reach their maximum market volume. In these cases, the current² number of products in use (or installed capacity for electricity generation technologies) is taken as a proxy for the peak market size. This is discussed further in Sections 4.2 and 5.1. In all cases a variety of alternative definitions were explored in the research, including when replacement products overtook sales of incumbent products, or when electricity production reached a fraction of total electricity sales (Hanna et al., 2015). The definitions above were chosen in order to maximise consistency across our categories of innovative product. In the case of electricity generation we chose fraction of total installed capacity of the technology in question (wind, gas, solar, etc.) rather than total installed capacity, electricity sales, or share of customers for the reason that in many (almost all) countries a mix of power generation sources are used and there is no reason to expect any one technology will come to provide all electricity. In line with the pragmatic approach taken generally we were seeking a measure that would permit a consistent and readily understood definition of widespread commercialisation.

We have characterised the innovation journey in terms of the cumulative adoption of each technology against time. Progress with innovation could alternatively be measured through patent applications or technological performance and there are alternatives to using time as a metric for the innovation journey, such as tracking innovation against investment in technological development (Taylor and Taylor, 2012). However the focus of the paper is on the temporal dimension of technology development.

In Appendices B and C we present in further detail the timelines for each technology reviewed and some of the assumptions and judgements that were used to determine these timescales. We also provide the historical context and reference sources used to construct timelines for each technology.

A final and crucially important point of note before discussing specific technologies is that innovation does not proceed in a linear fashion from one stage of the technological diffusion to the next (Wilson and Grubler, 2014). The 'innovation journey' should not be conflated with any suggestion that R&D is 'finished' for any particular technology. Technologies rarely move out of the R&D stage completely following market introduction. Ongoing R&D is crucial to improving performance and reducing costs even in the most mature of technologies (Sanden and Azar, 2005). This is a fundamentally important caveat to the historical analyses that follow and a point we reiterate in our discussion and conclusions.

4. Key findings on innovation timescales

4.1. Introduction

This section provides an empirical review of 13 products and technologies whose journeys from invention to widespread commercialisation span the late 19th Century through to the present time. A detailed review of sources of historical information is drawn together with the definitions of the start of each stage and reaching of widespread commercialisation, as explained in Section 3.2. Technologies are organised into three categories, described below in order to inform a discussion of factors which affect the time taken for products to enter into markets and reach widespread commercial use.

4.2. Chronology of innovations

The technologies here represent a diverse range of consumer goods, together with electricity generation options. Most were invented after 1940 – while the car and cathode ray tube TV (CRT TV) were invented in 1885 and 1912 respectively (Fig. 4). More recent products appear on the whole to have developed and commercialised more quickly, although additional research based on a larger sample of technologies more evenly distributed through time would be needed to confirm this trend. The nature of our definition of widespread commercialisation also creates a tendency towards comparatively longer timescales for older, obsolete products such as the CRT TV, whose market volume peaked and then declined. Conversely, our approach might create a tendency towards shorter timescales for those products which can be considered widely commercialised, but may have yet to reach their ultimate market volume. This latter effect applies both to the car and to more recent products in the sample (see Section 5.1).

 $^{^{2}}$ Based on the latest available year of data collected as part of the evidence review.



Fig. 4. Historical timelines of innovation for all technologies reviewed.



Fig. 5. Duration of innovation for all technologies reviewed.

4.3. Duration of innovation: general observations

The findings demonstrate that the journey to our definition of widespread commercialisation takes between two and four decades for most of the products and technologies reviewed. The median time taken from invention to widespread commercialisation was 32 years. Fig. 5 orders the innovations by the duration of their innovation journey. The shortest time to commercialisation was 20 years (for the LCD TV), while the longest was 69 years (the car). Although the median duration of the two innovation stages (invention, development and demonstration, and market deployment and commercialisation) is the same at 18 years, the length of each phase shows considerable variation across the innovations reviewed. For example, the shortest time taken for invention, development and demonstration was 5 years (the compact fluorescent light bulb and the LCD TV), while the longest was 37 years (solar PV).

Similarly, the shortest time from market introduction to widespread commercialisation was 8 years for the VCR whereas the longest was 46 years (the car). In addition, CCGT in the UK took just 5 years to reach widespread commercialisation after a 15-year moratorium on gas generation and during the 'dash for gas.' The particular case of CCGT is discussed in Section 5.1.

4.4. Duration of innovation by technology category

Fig. 6 groups the innovation timescales by technology category: Novel energy end use and consumer products (new products with entirely new markets); Replacement energy end use and consumer products (new technologies which replace but perform a similar function to an existing product); and electricity generation technologies.



Fig. 6. Historical timelines by product / technology category.

Table 5

Range of innovation duration by technology category (median durations indicated in brackets).

Innovation group	Innovation sub-category	Invention, development and demonstration	Market deployment and commercialisation	Total duration (years)
Energy end use and consumer	Novel products for new markets	7–26 (21)	8–46 (16.5)	22–69 (35.5)
products	Replacement products	5-13 (10.0)	14-27 (15.0)	20-32 (27.0)
Energy end use or supply	Energy end use and consumer products	5–26 (12.0)	8–46 (15.0)	20–69 (27.0)
	Electricity generation technologies	18–37 (24.0)	18–21 (19.0)	39–55 <i>(43.0)</i>
All innovations	Median	18.0	18.0	32.0
	Minimum	5	8	20
	Maximum	37	46	69

Table notes.

Novel products for new markets: Cars; Cathode Ray Tube (CRT) TV; ATM/Cash cards; and VCR.

Replacement products: Mobile phone; Thin Film Transistor Liquid Crystal Display (TFT-LCD) TV; lithium ion rechargeable batteries; compact fluorescent light bulbs (CFLs); and LED lights.

Electricity generation technologies: CCGT, nuclear power, wind electricity and solar PV.

The market deployment and widespread commercialisation phase and the median total timescale for the electricity generation technologies include the EU regulatory restriction on gas generation (and therefore CCGT) from 1975 to 1990.

5. Discussion

In what follows we consider the implications and limitations of the findings of the review, and compare the findings to the observations made by other authors. Overall, electricity generation technologies exhibit some of the longest commercialisation time periods within the technologies reviewed. The median time from invention to widespread commercialisation was 43 years for the four electricity generation technologies included in our study. In comparison, the nine energy end use and consumer products have a median duration of 27 years from invention to widespread commercialisation (Table 5). Lund (2006) also finds a shorter time to market penetration for some energy end use products (which can be less than 25 years) compared to energy production technologies. Our research does not provide a clear reason for this difference; however it is consistent with a number of potential explanations. For one, electricity generation technologies typically have long asset lives - several decades in the case of power stations. They therefore take longer to replace than many consumer products (Grubler, 1998). A number of studies also indicate that electricity generation technologies require time to achieve up-scaling, thereby enabling the widespread deployment of large-scale units (Bento and Fontes, 2015; Grubler, 2012; Wilson, 2012).

The two renewable generation technologies, solar PV (55 years in Germany) and wind (40 years in Denmark), also have amongst the longest timescales of the 13 innovations included in our study. Slow diffusion timescales for renewable energy technologies were observed in the early 2000s in European countries by Negro et al. (2012), who point to a range of innovation system failures which constrain their deployment, such as a lack of long term and consistent institutional support. Such failures are compounded by lock-in to incumbent fossil fuel technologies which have become optimally aligned with supporting institutions and have benefited from economies of scale and technological learning over extensive periods of time (Negro et al., 2012). However it is important to note the significant growth in renewables capacity experienced in many countries in more recent years. It is possible that diffusion rates will increase as technology costs fall.

As noted in Section 3.1, it has been suggested that new products which replace the function provided by a pre-existing product diffuse more quickly than those products which provide a completely new service (Grübler et al., 1999). If new institutions or infrastructures are needed to commercialise a technology, it will tend to take a longer time to develop and its diffusion is likely to be slower. Conversely, the rate of technological diffusion may be faster for replacement products which have a greater perceived relative advantage over incumbent or rival products (Grubler, 1998). More rapid formative phases have been observed for innovations which are 'ready substitutes' for pre-existing technologies, compared to 'non-ready substitutes' which take longer to align with new institutions and stimulate market demand (Bento and Wilson, 2016). This is broadly consistent with our review findings. which show that replacement consumer products had a shorter median period from invention to widespread commercialisation (27 years) than products aimed at entirely new markets (35.5 years) (Table 5). Many factors are relevant here, not least the dramatic reduction in retail prices of 'aspirational' products such as flat screen TVs relative to prices for incumbent products and to incomes (Energy Saving Trust, 2007). The LCD TV travelled from invention to widespread commercialisation in less than half the time (20 years) it took the Cathode Ray Tube (CRT) TV (44 years).

It is important to exercise considerable caution in drawing overly firm conclusions from the relatively small sample the review considered. Moreover, two of the four new products - cash cards/ATMs and the videocassette recorder – achieved widespread commercialisation in less than three decades (Fig. 5). Nevertheless, it appears likely that replacement products which do not require new infrastructure or institutions will tend to reach widespread commercialisation more rapidly than entirely new concepts that require wider system change.

5.1. Limitations

In seeking a widely applicable definition of widespread commercialisation, based upon available data, our analysis gives rise to a number of difficulties and limitations. Some of these manifest with regards to particular technologies, others are more generic. We discuss each in turn.

The car: The long timescale for the car came as a considerable surprise given the rapid adoption of automobiles in the US in the early twentieth century. The principal reason it took so long for the car to reach our definition of widespread commercialisation in the US is that extensive growth in the market for automobiles continued over most of the twentieth century. This was driven as much by population and economic growth, and growth in private incomes, as innovation in vehicle manufacturing and technology. Incomes needed to reach a level sufficient for widespread uptake of the car to be possible (Grubler, 1998), since the purchase of a car required (and still does) a very large expenditure relative to income compared to all the other consumer products reviewed.

Gas fired power stations: CCGT in the UK is also a special case, because it experienced a significantly later market introduction in the UK than globally (the first 'large' CCGTs over 100 MW were sold in Japan and France in 1970), due in part to an EU regulatory restriction on natural gas generation from 1975 to 1990 (Watson, 1997). The rapid deployment of CCGT that followed in the early 1990s has been associated with the liberalisation of the UK electricity market, and the so-called 'dash for gas' (Watson, 1997).

LED Lights: The replacement products selected in this study have gone on to replace the incumbent product(s) by achieving the largest share of annual sales or products in use, with one exception – the LED light bulb. LED lights have considerable potential to replace incumbent lighting products and in particular the CFL, but are still in an early stage of market growth and represent only a small fraction of light bulbs installed in UK households: 0.6% in 2016 (BEIS, 2017b). For these reasons, LEDs have not yet reached widespread commercialisation according to our definition, despite their widespread availability to consumers.

More broadly it is likely that as well as technology or market specific factors the length of any technology's gestation period could be affected by macroeconomic and societal factors, from wars and recessions to changing use of digital and social media. In compiling evidence for our case studies, we note for instance that the expansion of car manufacturing in the US in late 1920s was disrupted by the Great Depression (Mercer and Douglas Morgan, 1971). Bento and Wilson (2016) show a relatively quick formative phase for nuclear power associated with the unusually high demand pull from military users during World War Two.

The findings on innovation timescales are sensitive to the technology case studies selected. The approach taken to selecting technological innovations in our study sought to allow a comparison of innovation timescales across a diversity of energy-related and broader consumer products, so that insights might be drawn from observed differences between consumer products and energy supply or end use technologies. Our evidence review has not set out to categorise technological innovations or products in terms of whether they are consumer, investor or government-led, or to consider the impact of these aspects on innovation timescales. However, technology and market factors influencing commercialisation timescales may vary considerably between consumer-led products and those driven more by policy or investment by manufacturers or utilities. For example, compared to consumer-led products (e.g. the mobile phone or LCD TV), electricity generation technologies such as solar PV and wind may lack demand pull since their benefits are less immediately tangible to consumers and may be 'non-market' factors (such as lower emissions) which Grubler et al. (2016) associate with slow energy transitions.

It is conceivable that emerging developments such as the rollout of smart meters, household ICT and energy management, and peer-to-peer electricity trading could result in a blurring of traditional boundaries between electricity as a supply or end use activity. This could lead to increasing levels of consumer demand for smart and flexible energy services and potentially shorter timescales to widespread commercialisation for associated carbon abatement technologies.

A further caveat is with respect to the point of widespread commercialisation and peak product use. Widespread commercialisation has been taken as 20% of peak or current installed products in use (or installed capacity for electricity generation). Older technologies such as the CRT TV, which have already passed peak volume and declined into obsolescence, will tend to show a more extended timescale in comparison to products which have not yet reached their peak. For many of the more recent products in the sample as well as for the car, markets still appear to be growing. Hence the 20% level may be reached earlier, because the full extent of growth in the volume of products is not yet known.

6. Conclusions and policy implications

Our review suggests first and foremost that the process of innovation, as defined from the point of invention to the point of widespread commercialisation, can be a multi-decadal process – taking thirty or forty years for many of the technologies in this study. We also note that in many cases the scientific principles that were put to use in new technologies were established at an even earlier stage.

The first policy insight which follows is that any suggestion that the principal response to the challenge of climate change should be an intensive research effort to discover new, breakthrough low-carbon technologies needs to be treated with a considerable degree of caution. Such an approach risks taking too long to deliver the solutions needed

to address climate change, even assuming entirely new, breakthrough technologies can be found.

Care needs to be taken in extrapolating innovation timescales from the past, and historical contexts from early in the twentieth century will obviously differ from the future of technological development. It is of course possible that entirely new, low carbon, technologies could be discovered in the coming years and that these could be pushed through the various phases of research, demonstration and market introduction exceptionally quickly. However the attainment of a low-carbon energy system in the coming decades is far more likely to be the result of the full commercialisation of energy technologies already available, or which are close to being deployed at scale. This implies that technology deployment policies, including regulation, subsidy schemes and targets continue to be at least as important as support for fundamental research and development.

Whilst not the principal focus of our research it is also important to reiterate that research and development continue throughout the period of commercialisation. Ongoing research is often undertaken in concert with the learning, innovation and cost reduction which results from the manufacture, market penetration and use of technologies. As such, the choice for policy makers is not one of *either* supporting R&D *or* supporting deployment, but of how to strike the right *balance* between these policies as technologies are first developed and then enter markets.

Another insight from this analysis concerns the potential for more rapid commercialisation of end-use products when compared to electricity generation technologies. The review does not provide a definitive view, since the sample size is small. However, the products reviewed here suggest that consumer products such as phones and flat screen TVs have tended to commercialise relatively rapidly. There is also evidence to suggest that replacement products which can benefit from established infrastructures and institutions commercialise more rapidly than those which are entirely new concepts.

A taxonomy of key low-carbon technologies that considers these factors could prove fruitful in identifying those which could penetrate into markets rapidly, versus those that are likely to do so more gradually. It may also be possible to begin to assess where various barriers to commercialisation exist and to address these directly.

Understanding the realistic timescales for the commercialisation of different energy technologies in this way could provide a critically important 'reality-check' for future low-carbon scenario-analysis, in terms of which technologies are required to be deployed at given points in time. Further research could investigate the consistency between historically-observed and modelled diffusion rates for different low carbon technologies, across a range of energy systems models and scenarios. This could help identify and avoid significant bottlenecks or errors, as well as identify where different policy efforts should be targeted to achieve the desired transition pathway.

As we look to the future it is important that innovation policy continues to recognise that sustained support for low carbon technologies is likely to be required over many decades. Policy efforts to promote 'breakthrough' technologies are important, particularly in areas where low carbon substitutes for existing products or processes are yet to be found. However it would be unwise if climate policy were to focus excessively on RD&D, neglecting the time and effort needed to ensure that low carbon technologies find a route to market. Climate policies will need to continue to ensure that lower carbon options are given the support that they may need to compete with incumbent technologies until they become established and secure widespread use.

Acknowledgements

The research presented in this paper was undertaken as part of the research programme of the UK Energy Research Centre - Phase 3, which is funded by the UK Research and Innovation Energy Programme (Grant reference EP/L024756/1). Part-funding for the original research was also provided by the Committee on Climate Change. We thank P. Pearson and J. Skea at Imperial College London, C. Wilson (University of East Anglia) and R. Fouquet (London School of Economics) for their expert input on the emerging findings. We are also grateful for the constructive comments provided by two anonymous reviewers.

Author contributions

RG proposed the research design. RH led the review and analysis of empirical evidence, with contributions from RG, PH, JS and AG. RG wrote the first draft of the paper. All authors contributed to the final version.

Competing financial interests

The authors declare no competing financial interests.

Appendix A. Keyword searches used to identify relevant literature in Science Direct and World Cat

Initial search term categories		Subsequent search term filters
Innovation	Temporal	Innovative technology or product
Innovation	Time	Cars / automobiles
Research	Life	"Catalytic converter"
Mass market	Cycle	"Lithium ion" AND "car batteries"
"Market saturation"	Rate	"Lithium ion" AND "rechargeable batteries"
Commercialisation	Speed	Television / "Cathode Ray Tube TV" / "CRT
Deployment	History	TV"/ "Liquid Crystal Display TV" / "LCD TV"
Diffusion	-	"Automatic Teller Machine" / ATM / "cash cards"
Uptake		"Videocassette recorder" / VCR
"Innovation life cycle"		Photocopier / "plain paper copier"
"Technology life cycle"		"Mobile phone"
Technology		"Compact fluorescent light bulbs" / CFLs
"Product development"		"LED light bulbs" / "LED lamps"
-		"Combined cycle gas turbines" / CCGT
		"Wind electricity"
		"Nuclear power"
		"Solar photovoltaics" / "solar DV"

(a) Novel products for	new markets – energy end use an	id consumer product	s	
Innovation	Geography for market introduction and widespread commercialisation*	Invention, development and demonstration	Market deployment and commercialisation (Widespread commercialisation taken at 20% of peak products in use or ownership)	Total time taken for widespread commercialisation (Time taken in years to reach 20% of peak products in use or ownership)
Cars Cathode Ray Tube (CRT) TV ATM/Cash cards Video cassette recorder (VCR)	US US UK UK	1885–1908 1912–1938 1964–1971 1956–1975	1908–1954 1938–1956 1971–1986 1975–1983	69 44 22 27
(b) Replacement produ	cts - energy end use and consum	er products		
Innovation	Geography for market introduction and widespread commercialisation*	Invention, development and demonstration	Market deployment and commercialisation (Widespread commercialisation taken at 20% of peak products in use or ownership)	Total time taken for widespread commercialisation (Time taken in years to reach 20% of peak products in use or ownership)
Mobile phone Thin Film Transistor Líquid Crystal Display (TFT-LCD) TV	US Global	1970–1983 1988–1993	1983–1999 1993–2008	29 20
Lithium ion rechargeable batteries for constructs recoducts	Giobal	1979–1991	1991–2006	27
Compact fluorescent light bulbs (CFLs)	UK	1976–1981	1981–2008	33
LED lights	UK	1992–2002	2002–2016**	24**
(c) Electricity generation	on technologies			
Innovation	Geography for market introduction and widespread commercialisation*	Invention, development and demonstration	Market deployment and commercialisation (Widespread commercialisation taken at 20% of peak installed capacity)	Total time taken for widespread commercialisation (<i>Time taken in years to reach 20%</i> of peak installed capacity)
Combined cycle gas turbines (CCGT)	UK	1949–1990	1990–1995	46
Wind electricity	Denmark	1957–1979	1979–1997	40
Nuclear power Solar photovoltaics	France Germany	1941–1959 1954–1991	1959–1980 1991–2009	39 55
(solar PV) – grid- connected				
*Congraphy is alobal for the	invention of all innovations raviawad			

R. Gross et al.

Appendix B. Comparison of the timescales, chronology and geographical scope of the data revealed through our review, upon which the discussion of findings in our paper draws

*Geography is global for the invention of all innovations reviewed. **This innovation has yet to reach the definition of widespread commercialisation to which the table column applies.

Appendix C. Summ each innovation rev	ary of the judgements made and the reference material used i	ı applying our definitions of invention, market iı	ntroduction and widespread commercialisation to
Novel products or	technologies for new markets		
Innovation	How year of invention / conception is defined	How year of first market introduction is defined	How year of widespread commercialisation is defined
Cars	First vehicle powered by a combustion engine was built in 1885 (Bellis, 2015).	The first production of the Ford Model T in 1908 (Dismukes et al., 2009) has been taken as the point of market introduction. This is consistent with Mercer and Douglas Morgan (1971), who state: The American automobile industry really began its commercial existence only in 1900, while the rise of national corporations and the first real growth of the industry did not start until after the short business recession of 1907.	In 2014 there were 258 million household vehicles in operation in the US, the highest number for any year from 1950 to 2014. 20% of this was achieved in 1954, when there were 53 million household vehicles in operation in the US (Oak Ridge National Laboratory, 2015).
Cathode Ray Tube (CRT) TV	Alan Archibald Campbell Swinton proposed an electronic television system in 1908. In 1912, Boris Rosing used a cathoderay tube (CRT) for electronic television display (Magoun, 2009).	The first electronic CRT television set was introduced to the US market in 1938. 'the DuMont Corporation was the first manufacturer to sell all- electronic television receivers to the public when it made its model 180 sets availablein the waning months of 1938' (Edgerton, 2007)	The peak number of CRT TVs in the US was 215–219 million in the mid-1990s. 20% of this is 43–44 million. The US had a total of 43 million TVs in use during 1956, with 47 million in 1957, so 20% of peak CRT TVs in use was reached in 1956, to the nearest year approximation (UNESCO, 1965; IEA, 2009. NationMaster. 2017a. MationMaster. 2017b.
Automatic teller machine (ATM)/Cash cards	The ATM / cash card represents a complex innovation which was developed based on the cumulative knowledge of a number of earlier innovations in the 1950s and 1960s, including self-service gas stations, automatic public transport ticketing and candy dispensers (Bátiz-Lazo, 2015). Nevertheless, the point of invention is taken as 1964, when Smiths Industries collaborated with Chubb & Sons Lock and Safe Company (Chubb) to develop a system for dispensing oil to tanker drivers using a punched card (Batiz-Lazo and Reid, 2008). This was the foundation upon which European banks conceived an automatic system to provide customers access to cash outside of bank opening hours. The first patent for a currency dispenser system was issued in 1966, integrating a private identification number (PIN) with a public number (PAN) (Batiz-Lazo and Reid,	The very first cash machine were installed by Barclays in London in 1967, while a Chubb ATM/ cash card system was also deployed in London a month later. Another cash dispensing machine was also installed independently in 1967, in the form of the Bankomat in Sweden (Batiz-Lazo and Reid, 2008; Bátiz-Lazo, 2015). However, it was not until 1971 that cash machines were more widely introduced to markets worldwide by manufacturers operating in Great Britain, the US and Japan, in their own countries as well as across Europe, Israel, Canada, and South America (Bátiz- Lazo, 2015).	Based on data on the number of installed ATMs in the UK, from 1975 to 2003 (Batiz-Lazo, 2007). The highest number of installed ATMs for any year was in 2003, when there were 46,461 ATMs. 20% of this was attained in 1986, in which year there were 10,330 installed ATMs in the UK (Batiz-Lazo, 2007).
	2008; Bátiz-Lazo, 2015)		

694

Video cassette recorder (VCR)	The date of invention has been taken as 1956, when the first practical magnetic tape video recorder, the VRX-1000, was released by Ampex (Magoun, 2009; Castonguay, 2006). This precedes the year attributed to the 'conception of the technical basis of the innovation', i.e. 1960 (Wonglimpiyarat, 2005).	The VCR was 'launched into the market-place' in 1975 (Wonglimpiyarat, 2005). This is the year when Sony released Betamax, with JVC's VHS following in 1976, and Phillips releasing the V2000 in 1978 (Castonguay, 2006).	The dataset used is sourced from DECC (2015) and includes annual totals for VCRs and DVDs together but not separately. The year with the highest number of VCRs (19.3 million) owned by UK households has been taken as 1997, a year before DVDs were introduced in the UK (Andrews, 1998). The first year in which the number of VCRs in UK households first reached 20% of 19.3 million was 1983 (DECC, 2015).
Replacement prod	ucts or technologies		
Innovation	How year of invention / conception is defined	How year of first market introduction is defined	How year of widespread commercialisation is defined
Mobile phone	First presentation of the mobile telephony concept by AT&T Bell Labs in the early 1970s (Yeo et al., 2015).	In 1983, the analogue mobile phone was launched commercially in the US by Motorola (Giachetti and Marchi, 2010).	Over the period 1975–2015, the maximum number of mobile phone subscriptions in the US was 377 million in 2015. This subscription total was calculated by multiplying 1.2 subscriptions per 100 people from World Bank (2017) by the US population of 321 million in 2015 (US Census Bureau, 2016). 20% of 377 million subscriptions was reached in 1999, when there were 83 million subscriptions (0.25 subscriptions per person multiplied by 273 million population) (World Bank, 2017, US Census Bureau, 2000). Product replacement: In 2002 globally, the number of mobile phone users overtook the 1.1 billion people with fixed telephone lines (Bohlin et al, 2010).
Thin Film Transistor Liquid Crystal Display (TFT- LCD)	The invention point has been taken as 1988, when Sharp developed a 14-in., colour TFT-LCD TV (Sharp, 2015).	The first LCD and plasma TVs were introduced to the market in the 1990s (IEA, 2009). Market introduction has been taken as 1993 based on Figure 97 in IEA (2009), which estimates global market share of annual sales by television technology type from 1990 to 2008.	Using global LCD TV shipment data (Statista, 2017, TrendForce, 2017) and varying the assumed average in-use life span of an LCD TV from 3 years to 10 years to derive total numbers in-use year-on- year shows peak number of products in-use occurring in 2016 (the most recent year of data). The peak volume is 655 million to 1.8 billion products-in use depending on the assumed product lifespan. 20% of this peak was reached between 2007 and 2009, varying by the assumed product lifespan. We have used 2008 as the mid-point. <u>Product replacement</u> : Sales of LCD TVs first overtook sales of CRT TVs in 2006 in the UK (Energy Saving Trust, 2007) and in 2007 in the EU (Fraunhofer Institute, 2007). The global market share of LCD TVs and CRT TVs was approximately equal in 2008 (IEA, 2009) while in 2010 LCD TVs comprised 77% of global TV shipments, with CRT TVs making up just 16% (Park et al., 2011).

products

Development of the lithium cobalt oxide positive electrode (Mizushima et al., 1980).

First commercially available lithium ion rechargeable battery (Sony, 1996).

Assuming an average battery lifetime of 2 years, the highest total capacity of lithium-ion batteries in use 2016. 20% of this peak consumption was reached in Assuming an average battery lifetime of 3 years, the rechargeable batteries has been estimated by using highest total capacity of lithium-ion batteries in use data on lithium-ion battery sales for consumer use lifetimes of 2 years and 3 years respectively based on (Muenzel et al., 2015) and (Tektronix, (n.d.)). in GWh from Deng (2015) and assuming battery for any year from 1998 to 2016 was 83 GWh in 2006, when total in-use capacity was 19.5 GWh. reached in 2006, when total in-use capacity was for any year from 1998 to 2016 is 119 GWh in 2016. 20% of this peak consumption was also The total in-use capacity of lithium-ion 23.8 GWh.

Product replacement: Global sales of lithium ion batteries reached 50% of all rechargeable battery sales in 1998 (Goonan, 2012).

achieved a dominant market share in UK homes in LED light bulbs are still at a relatively early stage of with only a small market share of installed lighting Product replacement: Energy saving light bulbs first incumbent lighting products to a significant extent, households in 2016, although LED lights comprised 'energy saving light bulbs' in UK households. 20% LED light bulbs have not yet reached 'widespread The maximum number of products for any given 2008, when there were 88 million energy saving appliances currently. We therefore consider that year was in 2016, when there were 418 million of this maximum was reached and surpassed in 2011, comprising 44% of all light bulb/lighting just 0.6% of all lighting appliances fitted in UK There were 4.9 million LED light bulbs in UK light bulbs in UK households (BEIS, 2017b). commercialisation and have yet to replace commercialisation' as defined in this study appliances in households (BEIS, 2017b). sales in 1998 (Goonan, 2012). households (BEIS, 2017b).

Globally in 2011, residential lighting accounted for 40% of the general lighting market, while LED lighting itself was estimated to have a 7% market share in residential lighting (McKinsey, 2012).

CompactIn 1976 Jan Hasker (Philips) developed the 'RecombinantfluorescentStructure CFL', while Edward Hammer (General Electric)light bulbsdeveloped the 'Spiral' CFL (Smithsonian Institution, 2015)(CFLs)

deployment of the CFL is the UK, and therefore we

(Miller, 2012). However, our geography for the

Phillips SL, was released to the US market in 1980

The first compact fluorescent light bulb (CFL), the

have taken market introduction as the first year that CFLs had a > 0% share of all lighting appliances in

UK households, i.e. 1981 (BEIS, 2017b).

In 1992, a visible blue and green InGaN LED was developed by Market introduction based on the first year of that LED light bulbs had a > 0% share of all lighting appliances in UK homes (BEIS, 2017b). milestone leading to white LED lighting, with the first white LED introduced by Nichia in 1996 (Sanderson and Simons, 2014). Nichia, attaining 10% efficiency. The InGaN LED was a key LED lights

696

Electricity generat	ion technologies		
Innovation	How year of invention / conception is defined	How year of first market introduction is defined	How year of widespread commercialisation is defined
Combined cycle gas turbines (CCGT)	1949 – General Electric installed a 'fully-fired combined cycle turbine' in the US, in which a 3.5 MW gas turbine operated in conjunction with a 35 MW steam plant. Nevertheless, Brown Boveri had considered the possibility of CGT since installing their first industrial gas turbine in Switzerland in 1939, leading to their installation of a CGT in Luxembourg in 1956 (Watson, 1997).	1970 globally – the first 'large' CCGTs (greater than 100 MW) were sold by Mitsubishi for location in Japan and by Brown Boveri for installation in France. These achieved contemporary standards for most CCGTs, in that two thirds of the power output was supplied by the gas turbine, with the steam turbine providing the remaining third. Additionally, plant efficiency for these CCGTs was starting to exceed 40% (Watson, 1997). In the UK, the first orders for CCGT occurred during the liberalisation of the electricity market and the 'dash for gas' from 1989/1990, with the first operational CCGT capacity in 1991 (Watson, 1997). CCGT in the UK experienced a significantly later market introduction in the UK than globally, due in part to an EU regulatory restriction on natural gas	CGGT achieved a peak installed capacity of 33 GW in the UK in 2012 (BEIS, 2017a). 20% of this was reached in 1995, when there was approximately 8.5 GW of CGGT in the UK (Kern, 2012).
Wind electricity	In 1957, Johannes Juul completed the construction of a 200 kW wind turbine known as the 'Gedser-molle' and the father of modern wind turbines. This is because, as a three-blade, upwind turbine, it was effectively a prototype for wind turbines produced during the oil crises of the 1970s ((Jones and Bouamane, 2011, Danish Wind Industry Association, 2003). For example, in 1975 Nasa refurbished the Gedser wind turbine in order to extract measurement data for the US wind energy mooranne (Danish Wind Industry Association, 2003).	generation 1001 1970 to 1990 (watsoft, 1997). The first commercial wind turbine (a 30 kW turbine) was installed in Denmark in 1979 (Ministry of Foreign Affairs of Denmark, 2016). This is consistent with Danish Energy Agency (2009), which states that 'the first batch-produced Danish wind turbines from the late-1970s had an output of 22 kW' (Danish Energy Agency, 2009).	Peak installed capacity of wind power was 5.3 GW in 2017 (Danish Energy Agency, 2017). 20% of this was reached in 1997, when wind power in Denmark had a total installed capacity of 1.1 GW.
Nuclear power	The history of electricity generation from nuclear power can be traced back to the 1950s, with the basic principles of an energy- releasing process using nuclear fission being established by 1939. The point of invention has been taken as 1941, when a report by the MAUD committee in the UK first proposed the 'use of uranium as a source of power' although this proposal was shelved muth the end of World War Two (WNA, 2014)	The first commercial nuclear reactors in France operated in 1959, while a pre-commercial reactor using a gas-graphite design was started up in France in 1956 (WNA, 2014).	Peak installed capacity of nuclear power was reached in France in 1999 when there was 63 GW (WNA, 2017). 20% of this is 12.6 GW. Installed nuclear capacity reached 8.6 GW in 1979 and 15 GW in 1980 (WNA, 2017), so the 20% threshold was crossed in 1980.
Solar photovoltaics (solar PV) – grid-connected	In 1954, three scientists at Bell Laboratories (Pearson, Chapin and Fuller) developed a laboratory-scale, silicon-based solar cell which could be translated rapidly to practical use, as a result of using silicon rather than selenium. In 1955, Bell Laboratories constructed a silicon solar PV module for outdoor use to power telephone lines in Georgia, with various commercial applications following (Perlin, 1999; Green, 2005).	The first large-scale diffusion project for grid-connected PV is considered to be the 1000 roofs programme in Germany (Brown and Hendry, 2009). This was initiated as a 'demonstration cum market formation programme' (Brown and Hendry, 2009) which forestalled parliamentary pressure for Feed-in-Tariffs, and led to 2100 installations with a total capacity of 5.3 MWp on private homes from 1991 to 1995. The scheme offered capital grants of 70% of the installation and capital costs, and by this time there was already a Feed-in-Tariff law in Germany (offering a rate of 90% of the electricity price) (Brown and Hendry, 2009).	Peak installed capacity of solar PV eligible under the German EEG (Erneuerbaren-Energien-Gesetz) was 39.3 GW in 2015 (Federal Network Agency, 2017). 20% of this peak capacity is 7.9 GW and this threshold was crossed in 2009, when there was 10.6 GW of installed capacity of PV in Germany (Federal Network Agency, 2017).

References

Allen, J.A., 1967. Scientific Innovation and Industrial Prosperity. Elsevier, Amsterdam.

- Anderson, D., Clark, C., Foxon, T., Gross, R., Jacobs, M., 2001. Innovation and the environment: challenges and policy options for the UK. London, UK.: Centre for Energy Policy and Technology (ICEPT) and the Fabian Society.
- Andrews, S., 1998. Testing the waters in Britain. Billboard, 13 June 1998.
- Arapostathis, S., Carlsson-Hyslop, A., Pearson, P.J.G., Thornton, J., Gradillas, M., Laczay, S., Wallis, S., 2013. Governing transitions: cases and insights from two periods in the history of the UK gas industry. Energy Policy 52, 25–44.
- Bartholomew, D., 1973. Stochastic Models for Social Processes. Wiley, New York.
- Bass, F., 1969. A new product growth model for consumer durables. Manag. Sci. 15, 215–227.
- Bassi, S., Duffy, C., Rydge, J., 2013. Decarbonising electricity generation [Online]. Grantham Research Institute on Climate Change and the Environment Available: http://www.lse.ac.uk/GranthamInstitute/publication/decarbonising-electricity-generation/> (Accessed 14 April 2016).
- Batiz-Lazo, B., 2007. Emergence and evolution of proprietary ATM networks in the UK, 1967–2000 [Online]. Munich Personal RePEc Archive. Available: https://mpra.ub.uni-muenchen.de/3689/1/MPRA_paper_3689.pdf (Accessed 14 April 2016).
- Bátiz-Lazo, B., 2015. A brief history of the ATM [Online]. Available: http://www.theatlantic.com/technology/archive/2015/03/a-brief-history-of-the-atm/388547/> (Accessed 3 December 2015).
- Batiz-Lazo, B., Reid, R. (2008). Evidence from the patent record on the development of cash dispensing technology. MPRA Paper No. 9461.
- Beis, 2017a. Digest of United Kingdom Energy Statistics. Chapter 5: Electricity. 5.7 Plant capacity - United Kingdom. [Online]. Department for Business, Energy & Industrial Strategy. Available: https://www.gov.uk/government/uploads/system/uploads/ attachment_data/file/633779/Chapter_5.pdf> (Accessed 6 August 2017).
- Beis, 2017b. Energy Consumption in the UK (ECUK) 2017 Data Tables [Online]. Department for Business, Energy & Industrial Strategy. Available: https://www.gov.uk/government/statistics/energy-consumption-in-the-uk (Accessed 6 August 2017).
- Bellis, M., 2015. The History of the Automobile [Online]. Available: http://inventors.about.com/library/weekly/aacarsgasa.htm> (Accessed 3 December 2015).
- Bento, N., Fontes, M., 2015. The construction of a new technological innovation system in a follower country: wind energy in Portugal. Technol. Forecast. Soc. Change 99, 197–210.
- Bento, N., Wilson, C., 2014. Formative phase lengths for a sample of energy technologies using a diverse set of indicators [Online]. International Institute for Applied Systems Analysis. Available: http://www.iiasa.ac.at/web/home/about/whatisiiasa/ informationkit/brief.html (Accessed 13 April 2016).
- Bento, N., Wilson, C., 2016. Measuring the duration of formative phases for energy technologies. Environ. Innov. Soc. Transit. 21, 95–112.
- Bergek, A., Tell, F., Berggren, C., Watson, J., 2008. Technological capabilities and late shakeouts: industrial dynamics in the advanced gas turbine industry, 1987–2002. Ind. Corp. Change 17, 335–392.
- Bohlin, A., Gruber, H., Koutroumpis, P., 2010. Diffusion of new technology generations in mobile communications. Inform. Econ. Policy 22, 51–60.
- Breakthrough Energy Coalition, 2016. The Principles [Online]. Available: http://www.breakthroughenergycoalition.com/en/index.html (Accessed 8 March 2018).
- Brown, J., Hendry, C., 2009. Public demonstration projects and field trials: accelerating commercialisation of sustainable technology in solar photovoltaics. Energy Policy 37, 2560–2573.
- Carter, C.F., Williams, B.R., 1957. Industry and Technical Progress: Factors Governing the Speed of Application of Science. Oxford University Press, London.
- Castonguay, 2006. 50 years of the video cassette recorder [Online]. Available: http://www.wipo.int/wipo_magazine/en/2006/06/article_0003.html (Accessed 3 December 2015).
- Ccc, 2013. Fourth Carbon Budget Review. Committee on Climate Change, London.
- Ccc, 2015. The fifth carbon budget The next step towards a low-carbon economy [Online]. Committee on Climate Change. Available: https://www.theccc.org.uk/publication/the-fifth-carbon-budget-the-next-step-towards-a-low-carbon-economy/ (Accessed 14 April 2016).
- Danish Energy Agency, 2009. Wind turbines in Denmark [Online]. Danish Energy Agency, Copenhagen (Available). http://www.ens.dk/sites/ens.dk/files/ dokumenter/publikationer/downloads/wind_turbines_in_denmark.pdf) (Accessed 23 March 2016).
- Danish Energy Agency, 2017. Overview table of wind power plants. Master data for wind turbines as at end of June 2017 [Online]. Available: https://ens.dk/service/statistik-data-noegletal-og-kort/data-oversigt-over-energisektoren (Accessed 6 August 2017).
- Danish Wind Industry Association, 2003. The Wind Energy Pioneers: The Gedser Wind Turbine [Online]. Available: http://xn-drmstrre-64ad.dk/wp-content/wind/miller/ windpower%20web/en/pictures/juul.htm> (Accessed 3 December 2015).
- Davies, S., 1979. The Diffusion of Process Innovations. Cambridge University Press, Cambridge.
- Decc, 2012. DECC Science and Innovation Strategy. Department of Energy and Climate Change, London.
- Decc, 2015. Energy consumption in the UK. Domestic data tables. Table 3.12: Number of appliances owned by households in the UK 1970 to 2014 [Online]. Department of Energy & Climate Change. Available: https://www.gov.uk/government/statistics/ energy-consumption-in-the-uk (Accessed 2 December 2015).
- Dechezleprêtre, A., Martin, R., Bassi, S., 2016. Climate change policy, innovation and growth [Online]. Grantham Research Institute on Climate Change and the Environment Available: http://www.lse.ac.uk/GranthamInstitute/wp-content/

uploads/2016/01/Dechezlepretre-et-al-policy-brief-Jan-2016.pdf> (Accessed 13 April 2016).

- Deng, D., 2015. Li-ion batteries: basics, progress, and challenges. Energy Sci. Eng. 3, 385-418.
- Dismukes, J.P., Miller, L.K., Bers, J.A., 2009. The industrial life cycle of wind energy electrical power generation: ARI methodology modeling of life cycle dynamics. Technol. Forecast. Social. Change 76, 178–191.
- Edgerton, G., 2007. The Columbia History of American Television. Columbia University Press, New York, US.
- Elsevier, 2015. Science Direct [Online]. Available: http://www.sciencedirect.com/ (Accessed 15 July 2015).
- Energy Saving Trust, 2007. The ampere strikes back: how consumer electronics are taking over the world. London.
- Faiers, A., Neame, C., 2006. Consumer attitudes towards domestic solar power systems. Energy Policy 34, 1797–1806.
- Federal Network Agency, 2017. Development of the installed capacity of installations eligible under the EEG by technology type, 2003–2015 [Online]. Available: {https:// www.bundesnetzagentur.de/DE/Sachgebiete/ElektrizitaetundGas/Unternehmen_ Institutionen/ErneuerbareEnergien/ZahlenDatenInformationen/zahlenunddatennode.html> (Accessed 6 August 2017).
- Fouquet, R., 2010. The slow search for solutions: lessons from historical energy transitions by sector and service. Energy Policy 38, 6586–6596.
- Foxon, T., 2003. Inducing Innovation for a Low-Carbon Future: Drivers, Barriers and Policies – A Report for The Carbon Trust. The Carbon Trust, London.
- Foxon, T., Gross, R., Chase, A., Howes, J., Arnall, A., Anderson, D., 2005. Innovation systems for new and renewable energy technologies: drivers, barriers and system failures. Energy Policy 33, 2123–2137.
- Fraunhofer Institute, 2007. EuP Preparatory Studies Lot 5: Televisions. Final Report on Task 2: Economic and Market Analysis. Berlin: Fraunhofer Institute for Reliability and Microintegration and Öko-Institut.
- Fri, R., 2003. The role of knowledge: technological innovation in the energy system. Energy J. 24, 51–74.
- Gao, L., Porter, A.L., Wang, J., Fang, S., Zhang, X., Ma, T., Wang, W., Huang, L., 2013. Technology life cycle analysis method based on patent documents. Technol. Forecast. Social. Change 80, 398–407.
- Geels, F.W., 2002. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. Res. Policy 31, 1257–1274.

Geroski, P.A., 2000. Models of technology diffusion. Res. Policy 29, 603–625.

- Giachetti, C., Marchi, G., 2010. Evolution of firms' product strategy over the life cycle of technology-based industries: a case study of the global mobile phone industry, 1980–2009. Bus. Hist. 52, 1123–1150.
- Gold, B., Rosegger, G., Perlman, M., 1984. Technological progress and industrial leadership: the growth of the U.S. steel industry, 1900–1970, Massachusetts Lexington Books.
- Goonan, T., 2012. Lithium use in batteries. U.S. Geological Survey Circular 1371. Virginia, US: U.S. Geological Survey.
- Grad, P., 2006. Biofuelling Brazil: an overview of the bioethanol success story in Brazil. Refocus 7, 56–59.
- Green, M.A., 2005. Silicon photovoltaic modules: a brief history of the first 50 years. Prog. Photovolt. Res. Appl. 13, 447–455.
- Greenhalgh, T., Peacock, R., 2005. Effectiveness and efficiency of search methods in systematic reviews of complex evidence: audit of primary sources. BMJ: Br. Med. J. 331, 1064–1065.

Gross, R., Stern, J., Charles, C.H.R.I.S., Nicholls, J., Candelise, C., Heptonstall, P., Greenacre, P., 2012. On Picking Winners: The Need for Targeted Support for Renewable Energy. Imperial College London and World Wildlife Fund, London.

- Grubler, A., 1998. Technology and Global Change. Cambridge University Press, Cambridge, UK.
- Grubler, A., 2012. Energy transitions research: insights and cautionary tales. Energy Policy 50, 8–16.
- Grübler, A., Nakićenović, N., Victor, D.G., 1999. Dynamics of energy technologies and global change. Energy Policy 27, 247–280.
- Grubler, A., Wilson, C., Nemet, G., 2016. Apples, oranges, and consistent comparisons of the temporal dynamics of energy transitions. Energy Res. Soc. Sci. 22, 18–25.
- Hanna, R., Gross, R., Speirs, J., Heptonstall, P., Gambhir, A., 2015. Innovation timelines from invention to maturity: a rapid review of the evidence on the time taken for new technologies to reach widespread commercialisation. UKERC Technology and Policy Assessment Working Paper. London: UK Energy Research Centre.
- Helm, D., 2010. Government failure, rent-seeking, and capture: the design of climate change policy. Oxf. Rev. Econ. Policy 26, 182–196.
- Helm, D., 2017. Cost of energy review. Available: https://www.gov.uk/government/publications/cost-of-energy-independent-review (Accessed 5 September 2018).
- Iea, 2000. Learning Curves for Energy Technology Policy. International Energy Agency, Paris, France.
- Iea, 2009. Gadgets and Gigawatts. Policies for Energy Efficient Electronics. Paris, France.
- Iea, 2015. Energy technology perspectives 2015 Mobilising innovation to accelerate climate action [Online]. International Energy Agency. Available: http://www.iea.org/etp/etp2015/> (Accessed 13 April 2015).
- Ipcc, 2015. Climate Change 2014: Mitigation of Climate Change: Working Group III Contribution to the IPCC Fifth Assessment Report.
- Iyer, G., Hultman, N., Eom, J., Mcjeon, H., Patel, P., Clarke, L., 2015. Diffusion of lowcarbon technologies and the feasibility of long-term climate targets. Technol. Forecast. Soc. Change 90, 103–118.
- Jones, G., Bouamane, L., 2011. Historical trajectories and corporate competences in wind energy [Online]. Harvard Business School. Available: http://www.hbs.edu/faculty/

Publication%20Files/11-112_05079f6f-9952-43fe-9392-71f3001ceae4.pdf (Accessed 3 December 2015).

- Kern, F., 2012. The development of the CCGT and the 'dash for gas' in the UK power industry (1987–2000). Final case study report as part of Work Package 2 of the UKERC project: 'CCS – Realising the Potential?'. London: UK Energy Research Centre.
- King, D., Browne, J., Layard, R., O'donnell, G., Rees, M., Stern, N., Turner, A., 2015. A global apollo programme to combat climate change [Online]. London School of Economics and Political Science. Available: http://cep.lse.ac.uk/pubs/download/special/Global_Apollo_Programme_Report.pdf (Accessed 13 April 2016). Kramer, G., Haigh, M., 2009. No quick switch to low-carbon energy. Nature 462,
- 568–569.
- Lund, P., 2006. Market penetration rates of new energy technologies. Energy Policy 34, 3317–3326.
- Magoun, A., 2009. Television: The Life Story of a Technology. Johns Hopkins University Press, Baltimore, US.
- Mansfield, E., 1965. Innovation and technical change in the railroad industry. In: NBER (Ed.), Transportation Economics. National Bureau of Economic Research.
- Mcdowall, W., 2015. Are scenarios of hydrogen vehicle adoption optimistic? A comparison with historical analogies. Environmental Innovation and Societal Transitions.
- Mckinsey, 2012. Lighting the way: perspectives on the global lighting market. [Online]. Available: http://www.led-professional.com/business/reports/lighting-the-way-perspectives-on-the-global-lighting-market-second-edition-2012-executive-summary-mckinsey/McKinsey%20-%20Company%20-%20Lighting%20the%20Way%202012.pdf; (Accessed 3 December 2015).
- Mercer, L.J., Douglas Morgan, W., 1971. Alternative interpretations of market saturation: evaluation for the automobile market in the late twenties. Explor. Econ. Hist. 9, 269–290.
- Miller, P., 2012. A Brighter Idea: The Untold Story of the CFL. Electr. J. 25, 56-64.
- Ministry of Foreign Affairs of Denmark, 2016. A world-leader in wind energy [Online]. Available: http://denmark.dk/en/green-living/wind-energy/> (Accessed 23 March 2016).
- Mission Innovation, 2016. Mission Innovation: Accelerating the clean energy revolution. [Online]. Available: http://mission-innovation.net/> (Accessed 13 April 2016).
- Mizushima, K., Jones, P.C., Wiseman, P.J., Goodenough, J.B., 1980. LixCoO₂ (0 < x < -1): a new cathode material for batteries of high energy density. Mater. Res. Bull. 15, 783–789.
- Muenzel, V., DE Hoog, J., Brazil, M., Vishwanath, A., Kalyanaraman, S., 2015. A multifactor battery cycle life prediction methodology for optimal battery management. In: Proceedings of the Sixth International Conference on Future Energy Systems. Association for Computing Machinery, Bangalore, India.
- Napp, T., Bernie, D., Thomas, R., Lowe, J., Hawkes, A., Gambhir, A., 2017. Exploring the feasibility of low-carbon scenarios using historical energy transitions analysis. Energies 10, 116.
- Nationmaster, 2017a. Media Television receivers: Countries Compared [Online]. Available: http://www.nationmaster.com/country-info/stats/Media/Television-receivers> (Accessed 6 August 2017).
- Nationmaster, 2017b. Media Televisions per 1000: Countries Compared [Online]. Available: http://www.nationmaster.com/country-info/stats/Media/Televisions-per-1000) (Accessed 6 August 2017).
- Negro, S.O., Alkemade, F., Hekkert, M.P., 2012. Why does renewable energy diffuse so slowly? A review of innovation system problems. Renew. Sustain. Energy Rev. 16, 3836–3846.
- Nemet, G.F., Baker, E., 2009. Demand subsidies versus R&D: comparing the uncertain impacts of policy on a pre-commercial low-carbon energy technology. Energy J. 30, 49–80.
- Oak Ridge National Laboratory, 2015. Transportation energy data book Chapter 8. Household vehicles and characteristics. [Online]. Available: http://cta.ornl.gov/data/index.shtml> (Accessed 2 December 2015).
- Oclc, 2015. World Cat [Online]. Online Computer Library Center. Available: https://www.worldcat.org/) (Accessed 15 July 2015).
- O'neill, B., Grubler, A., Nakicenovic, N., Obersteiner, M., Keywan, R., Schrattenholzer, L., Toth, F., 2003. Planning for future energy resources. Science 300, 581.
- Oster, S., 1982. The diffusion of innovation among steel firms: the basic oxygen furnace. Bell J. Econ. 13, 45–56.
- Park, W., Phadke, A., Shah, N., Letschert, V., 2011. TV energy consumption trends and energy-efficiency improvement options [Online]. Berkeley National Laboratory. Available: http://eetd.lbl.gov/sites/all/files/lbnl-5024e.pdf).
- Pearson, P.J.G., Arapostathis, S., 2017. Two centuries of innovation, transformation and transition in the UK gas industry: where next? Proc. Inst. Mech. Eng., Part A: J. Power Energy (0957650917693482).
- Perez, C., 2002. Technological Revolutions and Financial Capital: The Dynamics of Bubbles and Golden Ages. Edward Elgar, Cheltenham, UK.
- Perlin, J., 1999. From Space to Earth: The Story of Solar Electricity. Earthscan.

Policy Exchange, 2011. Climate change policy - time for plan B [Online]. Available: http://www.policyexchange.org.uk/publications/category/item/climate-change-policy-time-for-plan-b (Accessed 13 April 2016).

Rogers, E., 1962. Diffusion of Innovations. The Free Press, New York.

- Sanden, B., Azar, C., 2005. Near-term technology policies for long-term climate targets economy wide versus technology specific approaches. Energy Policy 1557–1576. Sanderson, S.W., Simons, K.L., 2014. Light emitting diodes and the lighting revolution:
- the emergence of a solid-state lighting industry. Res. Policy 43, 1730–1746. Saviotti, P.P., Metcalfe, J.S., 1984. A theoretical approach to the construction of tech-
- nological output indicators. Res. Policy 13, 141–151. Schumpeter, J.A., 1934. The Theory of Economic Development. Harvard University

Press, Cambridge MA.

- Sharp, 2015. History of TV making for over 50 years and LCD for over 30 years [Online]. (Accessed 3 December 2015).
- Smil, V., 2010a. Energy Myths and Realities: Bringing Science to the Energy Policy Debate. US Rowman and Littlefield, Washington DC.
- Smil, V., 2010b. Energy Transitions: History, Requirements, Prospects. Santa Barbara, CA Praeger.
- Smithsonian Institution, 2015. Inventing six modern electric lamps [Online]. Available: in4">http://americanhistory.si.edu/lighting/20thcent/invent20.htm#>in4 (Accessed 2 December 2015).
- Sony, 1996. Genryu: Sony 50th Anniversary, Sony Corporation.
- Sovacool, B.K., 2016. How long will it take? Conceptualizing the temporal dynamics of energy transitions. Energy Res. Social. Sci. 13, 202–215.
- Speirs, J., Gross, R., Heptonstall, P., 2015. Developing a rapid evidence assessment (REA) methodology. A UKERC TPA technical document. [Online]. UK Energy Research Centre. Available: http://www.ukerc.ac.uk/programmes/technology-and-policy-assessment.html> (Accessed 4 August 2017).
- Statista, 2017. Global shipments of LCD TV sets from 2009 to 2015 (in million units) [Online]. Available: https://www.statista.com/statistics/273601/global-shipments-of-lcd-tv-since-2009/> (Accessed 6 August 2017).
- Taylor, M., Taylor, A., 2012. The technology life cycle: conceptualization and managerial implications. Int. J. Prod. Econ. 140, 541–553.
- Tektronix (n.d.). Lithium-Ion Battery Maintenance Guidelines [Online]. Available: <<u>http://www.newark.com/pdfs/techarticles/tektronix/LIBMG.pdf</u>> (Accessed 6 August 2017).
- Trendforce, 2017. TrendForce Reports Global LCD TV Shipments Grew 1.6% Annually in 2016; Hisense Narrowly Beat TCL to Take Third Spot in Ranking [Online]. Available: <<u>http://press.trendforce.com/press/20170202-2743.html</u>> (Accessed 7 August 2017).
- UNESCO, 1963. Statistics on radio and television, 1950–1960 [Online]. Available http://unesdoc.unesco.org/images/0003/000337/033739eo.pdf> (Accessed 7 August 2017).
- UNFCCC, 2015. Conference of the Parties twenty-first session. Adoption of the Paris Agreement. Proposal by the President Draft decision -CP.21. [Online]. Paris, 30 November to 11 December 2015: United Nations Framework Convention on Climate Change Available: http://unfccc.int/resource/docs/2015/cop21/eng/l09r01.pdf (Accessed 13 April 2016).
- US Census Bureau, 2000. Historical National Population Estimates: July 1, 1900 to July 1, 1999, Population estimates program, P.D. Washington D.C.
- US Census Bureau, 2016. Table 1. Monthly Population Estimates for the United States: April 1, 2010 to December 1, 2017 (NA-EST2016-01) In: DIVISION, P. (ed.). Washington D.C.
- Utterback, J., 1994. Mastering the Dynamics of Innovation: How Companies Can Seize Opportunities in the Face of Technological Change. Harvard Business School Press, Boston, USA.
- Van Der Zwaan, B.C.C., Rosler, H., Kober, T., Aboumahboub, T., Calvin, K.V., Gernaat, D.E.H.J., Marangoni, G., Mccollum, D., 2013. A cross-model comparison of global long-term technology diffusion under a 2 °C climate change control target. Clim. Change Econ. 4, 1–24.
- Van Sluisveld, M.A.E., Harmsen, J.H.M., Bauer, N., Mccollum, D.L., Riahi, K., Tavoni, M., Vuuren, D.P.V., Wilson, C., Zwaan, B.V.D., 2015. Comparing future patterns of energy system change in 2 °C scenarios with historically observed rates of change. Glob. Environ. Change 35, 436–449.
- Watson, J., 1997. Constructing Success in the Electric Power Industry: Combined Cycle Gas Turbines and Fluidised Beds. DPhil, University of Sussex.
- Wilson, A., Grubler, C., 2014. Energy technology innovation. In: GRUBLER, A., WILSON, C. (Eds.), Energy Technology Innovation: Learning from Historical Successes and Failures. Cambridge University Press, Cambridge, UK.
- Wilson, C., 2012. Up-scaling, formative phases, and learning in the historical diffusion of energy technologies. Energy Policy 50, 81–94.
- Wilson, C., Grubler, A., 2010. Lessons from the history of technology and global change for the emerging clean technology cluster. Backgr. Pap. World Econ. Social. Surv. (WESS) 2011.
- Wilson, C., Grubler, A., Bauer, N., Krey, V., Riahi, K., 2013. Future capacity growth of energy technologies: are scenarios consistent with historical evidence? Clim. Change 118, 381–395.
- Winskel, M., Anandarajah, G., Skea, J., Jay, B., 2011. Accelerating the development of energy supply technologies: the role of research and innovation. In: SKEA, J., EKINS, P., WINSKEL, M. (Eds.), Energy 2050: Making the Transition to a Secure Low-carbon System. Earthscan, London.
- Wna, 2014. Outline History of Nuclear Energy [Online]. World Nuclear Association London. Available: http://www.world-nuclear.org/info/Current-and-Future-Generation/Outline-History-of-Nuclear-Energy/ (Accessed September 2015).
- Wind: 2017. Country profiles nuclear power in France [Online]. World Nuclear Association. Available: http://www.world-nuclear.org/information-library/
- country-profiles/countries-a-f/france.aspx> (Accessed 6 August 2017). Wonglimpiyarat, J., 2005. Does complexity affect the speed of innovation? Technovation
- 25, 865–882. World Bank, 2017. Mobile cellular subscriptions (per 100 people) [Online]. Available:
- Yeh, S., 2007. An empirical analysis on the adoption of alternative fuel vehicles: the case of natural gas vehicles. Energy Policy 35, 5865–5875.
- Yeo, W., Kim, S., Park, H., Kang, J., 2015. A bibliometric method for measuring the degree of technological innovation. Technol. Forecast. Social. Change 95, 152–162.